# **@AGU**PUBLICATIONS



## **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1002/2017GL074443

#### **Key Points:**

- We constructed statistically significant power-function regression equations to estimate lake volume based on DEMs derived from images
- We found the different results in 1999 and 2008 through compared the water supply rate before and during a GLOF
- Quickly released intermediate- and short-term water storage by glaciers probably will increase the risk of GLOF

Supporting Information:

Supporting Information S1

Correspondence to: Y. Ding, dyj@lzb.ac.cn

#### Citation:

Shangguan, D., Ding, Y., Liu, S., Xie, Z., Pieczonka, T., Xu, J., & Moldobekov, B. (2017). Quick release of internal water storage in a glacier leads to underestimation of the hazard potential of glacial lake outburst floods from Lake Merzbacher in central Tian Shan Mountains. *Geophysical Research Letters*, 44, 9786–9795. https://doi.org/ 10.1002/2017GL074443

Received 8 JUN 2017 Accepted 15 SEP 2017 Accepted article online 18 SEP 2017 Published online 5 OCT 2017

©2017. American Geophysical Union. All Rights Reserved.

# Quick Release of Internal Water Storage in a Glacier Leads to Underestimation of the Hazard Potential of Glacial Lake Outburst Floods From Lake Merzbacher in Central Tian Shan Mountains

Donghui Shangguan<sup>1,2</sup> (D), Yongjian Ding<sup>1,3</sup> (D), Shiyin Liu<sup>4</sup> (D), Zunyi Xie<sup>5</sup> (D), Tino Pieczonka<sup>6</sup> (D), Junli Xu<sup>1,7</sup> (D), and Bolot Moldobekov<sup>8</sup>

<sup>1</sup>State Key Laboratory of Cryospheric Science, Northwest Institute of Eco-Environment and Resources, Chinese Academy of Sciences, Lanzhou, China, <sup>2</sup>Department of Geography, University of Zurich, Zurich, Switzerland, <sup>3</sup>University of Chinese Academy of Sciences, Beijing, China, <sup>4</sup>Institute of International Rivers and Eco-security, Yunnan University, Kunming, China, <sup>5</sup>Centre for Biodiversity and Conservation Science, School of Earth and Environmental Sciences, University of Queensland, St Lucia, Queensland, Australia, <sup>6</sup>Institute for Cartography, Technische Universität Dresden, Dresden, Germany, <sup>7</sup>Surveying and Mapping Department, Yancheng Teachers University, Yancheng, China, <sup>8</sup>Central-Asia Institute for Applied Geosciences, Bishkek, Kyrgyzstan

**Abstract** Glacial meltwater and ice calving contribute to the flood volume of glacial lakes such as Lake Merzbacher in the Tian Shan Mountains of central Asia. In this study, we simulated the lake's volume by constructing an empirical relationship between the area of Lake Merzbacher, determined from satellite images, and the lake's water storage, derived from digital elevation models. Results showed that the lake water supply rate before Glacial Lake Outburst Floods (GLOFs) generally agreed well with those during the GLOFs from 2009 to 2012 but not in 2008 and 2015. Furthermore, we found that the combination of glacial meltwater and ice calving is not enough to fully explain the supply rate during GLOFs in 1996 and 1999, suggesting other factors affect the supply rate during GLOFs as well. To examine this further, we compared the water supply rate before and during GLOF events in 1999 and 2008. We inferred that quickly released short-term and intermediate-term water storage by glaciers have likely contributed to both flood events in those years. This study highlights the need to improve our understanding of the supply component of outburst floods, such as irregularly released stored water may lead to GLOF events with generally three different types: case I (singular event-triggered englacial water release), case II (glacier melt due to temperature changes), and case III (englacial water release mixed with glacier melt).

**Plain Language Summary** Glacial lake outburst is one of natural hazards related to glaciers. The glacier lake outburst flood (GLOF) of Merzbacher which almost occurs every year threatens lives, livelihoods, and infrastructure both in the mountains and downstream, posing myriad challenges for the communities. The regional warming leads to the increasing to supply for the Lake Merzbacher from Inylchek glacier. Importantly, quickly released intermediate- and short-term water storage by glaciers probably contributed to floods, which increase flood volume and lead to severe risk on the downstream socioeconomy.

### 1. Introduction

A growing body of evidence suggested that glacial lakes develop and expand during periods of glacier recession, particularly in the context of contemporary climate change (Bolch et al., 2011; Wang et al., 2013; Yao et al., 2010). Glacial lakes are common in High Mountain Asia and some pose a significant risk of lake outburst floods, which are one type of jökulhlaup (a subglacial outburst flood) (Iwata et al., 2002; Janský et al., 2010; Richardson & Reynolds, 2000; Zhang et al., 2015). Lake Merzbacher (42°13'N, 79°52'E, about 3250 m above sea level (asl)) is a glacial lake located in the Kumarik Catchment, in the upper reaches of the Aksu River (Figure S1 in the supporting information). Lake Merzbacher is composed of a lower lake and an upper lake which are connected by the Merzbacher River (Glazirin, 2010). The lower lake is an ice-dammed lake (Häusler et al., 2016; Merzbacher, 1905) that has produced a total of 62 recorded outbursts between 1932 and 2008, most of which occurred in July, August, and September (Shen et al., 2009). Such Glacial Lake Outburst Floods (GLOFs) can cause great damage to infrastructure and human populations surrounding the glacial lakes, including destruction of bridges and damage to agricultural and industrial land

(Wortmann et al., 2014). Large floods also stress local disaster relief funds (Jongman et al., 2014). Therefore, a scientific assessment of the status of glacial lakes and of their potential to trigger outburst floods is critical for adaptation to these hazards.

The cause of these sudden releases of water from the lower lake is still not well understood; however, researchers believe that release of the lake's water begins when the ice dam created by the advancing southern lnylchek Glacier floats upward, thereby opening a system of englacial channels within the glacier and releasing the water they contain (Wortmann et al., 2014). Rising air temperatures and thinning of the ice dam created by this glacier appear to be the two main reasons for outburst floods, and the peak discharge  $(Q_{max})$  of the floods has increased as a result of climate warming (Glazirin, 2010; Ng et al., 2007).

Monitoring and assessment of potential glacial hazards are the key to their mitigation (Richardson & Reynolds, 2000). The timing, peak discharge ( $Q_{max}$ ), and flood volume ( $V_t$ ) of jökulhlaups are three key parameters that define a GLOF (Kingslake & Ng, 2013b; Liu, 1992; Ng et al., 2007) and are important for providing early warning of a flood and reduction of economic losses (Huggel et al., 2002; Osti et al., 2013). The duration of jökulhlaup flood discharges and the flood volume have been well simulated using numerical models (Kingslake & Ng, 2013b; Ng & Liu, 2009) and data from satellite images (Xie et al., 2013). However, despite these successes, it has proven difficult to predict GLOFs (e.g., GLOF in winter and GLOF with double peaks) that result from increasing temperatures, since these above mentioned approaches did not consider the englacial water storage.

Several methods can be used to estimate a glacial lake's storage volume, such as the stereophotogrammetry approach that has been used for Lake Merzbacher (Kuzmichenok, 1984); the development of an empirical formula for the maximum lake depth to simulate the volume of lakes in the Tian Shan Mountains (Konovalov, 1991), Swiss Alps (Huggel et al., 2002), and Bhutan (Osti et al., 2013); and the use of high-resolution echo sounder SONAR such as the SyQwest Hydrobox in the Tian Shan Mountains (Janský et al., 2010) and the Himalaya (Yao et al., 2012). The amount of water stored in and released from a glacial lake can also be used to calculate the peak discharge during an outburst flood and subsequent evolution of the flooding (Georgiou et al., 2009; Yao et al., 2012).

A numerical model that coupled a thermomechanically evolving subglacial channel, distributed cavity drainage, and basal sliding along a subglacial flood path was developed to model the flood discharge from Lake Merzbacher (Kingslake & Ng, 2013a). However, the model depended on an accurate assessment of the lake's volume, which was not available at the time of that study. Since the lake's basin is composed of ice, the subsurface ice melted and deformed as a result of regional warming, and these changes affected the behavior of the northern Inylchek Glacier (Sorg et al., 2012). The lake basin and lake volume both varied in response to these changes. Although the river discharge measured at the Xiehela hydrological station, 75 km downstream from the lake (Kingslake & Ng, 2013b; Ng et al., 2007), can be used to quantify the volume of released flood water, we lack adequate knowledge of the lake's volume (Wortmann et al., 2014). Field surveys are prohibitively difficult or expensive in most glacial areas and cannot be performed repeatedly at short intervals to assess changes in the lake area and volume, so a faster and more economical approach is urgently needed. Since there was relatively little water remained in Lake Merzbacher after a GLOF, we were able to survey the lake and determine its storage capacity by using ground mapping, remote sensing, and other tools (Fujita et al., 2013). Satellite observations with high-spatial resolution (e.g., SPOT-5 and ALOS) have been widely applied to obtain measurements of glacial topography (Berthier & Toutin, 2008; Uchiyama et al., 2008). Thus, remote sensing technique offers a promising approach that allows the development of a digital elevation model (DEM) that could be used to rapidly calculate lake volume. If satellite data with sufficient temporal resolution are available, this would potentially allow monitoring of the growth of lake volume by hypsometric analysis (Zhang et al., 2017) in real time and the prediction of GLOFs.

The initial lake volume, water supply rate, and water-equivalent calving flux (*c*) are three important factors that affect the flood volume (Ng & Liu, 2009). It is also useful to estimate the glacial meltwater supply to the lake during the flood event. In this study, we used the model of Ng et al. (2007) to estimate Lake Merzbacher's water supply. This model is based on the parameterization of a temperature index (see Text S1) (Ng & Liu, 2009). We defined the water-equivalent calving flux as a constant. No previous studies suggest that other factors contribute to the flood volume of Lake Merzbacher. However, the fact that the lake's outburst floods occur irregularly, sometimes even outside the expected summer period, indicates that other

supply mechanisms probably exist; therefore, we compared water supply rates before GLOFs and during GLOFs in several years. Through this work, we found inference of another englacial supply source.

#### 2. Data and Methods

Table S1 summarizes the satellite data that were the basis for our analyses of the lake's area. All DEMs were processed using version 2011 of the Leica Photogrammetric Suite (www.hexagongeospatial.com) (Shangguan et al., 2015) (Figure S6). The spatial resolutions of the DEMs (Table S1) were resampled to a common resolution of 30 m (by means of bilinear interpolation) based on the UTM WGS84 reference system. A coregistration to remove the horizontal and vertical offsets before differences analysis was used in this study (Nuth & Kääb, 2011). We used SPOT-5 high-resolution geometric DEM data combined with ALOS DEM data (to compensate for missing data in the two DEMs; see Text S2) to determine the lake's extent and to extract a DEM that would let us determine its storage capacity. We constructed a relationship between the lake's area and volume based on these data. Using the relationship, we calculated the lake's water supply rate and flood volume between 2008 and 2012 using a series of area changes derived from HJ1A/1B satellite images (Xie et al., 2013). We analyzed the water source for the flood volume and discussed possible causes of the flooding. The overall data analysis procedure is depicted by Figure S2 and the associated Text S2.

#### 2.1. Relationship Determination Between Lake Area and Volume

The water storage capacity was expressed using an empirical relationship between surface area and volume (Huggel et al., 2002; Zhang et al., 2017). To provide this relationship, we analyzed the DEM data from the following data sets: the 1974 KH-9 (earthexplorer.usgs.gov), the 2000 SRTM (http://www2.jpl.nasa.gov/srtm/), and the 2008 SPOT/2006 ALOS DEM (see supporting information for details). We calculated the projected area, surface area, and surface volume by using the surface volume tool in version 9.3 of ArcGIS (http:// www.esri.com/) relative to a given base height. We then generated an area-volume dependency curve (Figure 1). We constructed statistically significant power-function regression equations to estimate lake volume based on data from 1974 (Figure 1a), 2000 (Figure 1b), and 2008 (Figure 1c):

$$V_{1974} = 0.0153S^{1.1433} \tag{1}$$

$$V_{2000} = 0.0128S^{1.3584} \tag{2}$$

$$V_{2008} = 0.0160S^{1.4510} \tag{3}$$

where V represents the lake's volume (km<sup>3</sup>) in the indicated year and S represents the lake's surface area (km<sup>2</sup>). All three regressions were strong and significant ( $R^2 > 0.94$ , p < 0.05).

#### 2.2. Extraction of Initial Lake Volume, Water Supply Rate, and Flood Volume From 2008 to 2012

We obtained the total lake area from Xie et al.'s study, which was conducted from 2009 to 2011 (Xie et al., 2013). In addition, we added the lake areas in 2008 and 2012 derived from satellite data (Table S1). The initial lake volumes were calculated by using equations (1) to (3), with the lake area close to the time of a GLOF in each year as the input parameter. On this basis, the initial lake volume between 2008 and 2012 ranged from  $0.60 \times 10^8$  to  $1.19 \times 10^8$  m<sup>3</sup>. We calculated the mean water supply rate before a GLOF as follows (Ng & Liu, 2009):

$$Q_i = \frac{V_{i+1} - V_i}{t} \tag{4}$$

where  $Q_i$  represents the water supply rate (m<sup>3</sup> s<sup>-1</sup>) between times *i* and *i* + 1,  $V_i$  and  $V_{i+1}$  represent the volumes (m<sup>3</sup>) at times *i* and *i* + 1, and *t* represents the time interval between times *i* and *i* + 1.

#### 2.3. Hydrologic Parameters Obtained From Field Observations

Our hydrologic data came from records obtained from Xiehela Station, which is located downstream of the lake (Figure S1). The peak flood quantity passing through the mouth of the Inylchek glacier differs from the Xiehela gauging station (41°34'N, 79°37'E, 1487 m asl) (Glazirin, 2010). This is because the discharge recorded



**Figure 1.** The relationships between the area and volume of Lake Merzbacher based on the analysis of three sets of digital elevation model data: (a) KH-9, (b) SRTM, and (c) SPOT/ALOS. All regressions were statistically significant (p < 0.05).

at Xiehela Station represents the combination of two components: the base flow and the quick flow that results from the released water (Fan et al., 2013); Figure S3 provides an example from 2005. The recession of the quick flow that results from flooding is fast and shows a steep curve, whereas the recession of the base flow is slow and gentle because of the different flow processes that cause that flow. We therefore used inspection of the flow curves (e.g., Figure S4) to separate base flow from the postrelease quick flow (Ng et al., 2007). Ng et al. (2007) proposed that the water supply rate combines the effects of precipitation, glacial meltwater, and ice supplements derived from glacier kinematics. Melt rates were observed to vary from 2.8 to 6.7 cm d<sup>-1</sup> with a mean of 4.4 cm d<sup>-1</sup> in 2005 (Hagg et al., 2008). In comparison with glacier melt, the maximum precipitation was 28.9 mm before the flood peak of the GLOF (Table S2); hence, the water supply rate from precipitation was less than glacier meltwater. Furthermore, precipitation during spring and summer accounts for 90% annual precipitation and the share of solid precipitation increases with altitude in Kumarik catchment (Aizen et al., 1997; Kundzewicz et al., 2015). Consequently, streamflow in a heavily glaciated catchment like the Kumarik catchment is controlled by temperature at daily, monthly, and seasonal scales since precipitation is stored in glacier and snow areas (Kundzewicz et al., 2015).We expressed the mean water supply rate during a GLOF as follows:

$$Q_j = \frac{V_f - V_h}{t} \tag{5}$$

where  $Q_j$  represents the mean supply rate during the GLOF (m<sup>3</sup> s<sup>-1</sup>),  $v_f$  represents the flood volume observed at Xiehela Station (m<sup>3</sup>),  $V_h$  represents the initial lake volume (m<sup>3</sup>), and *t* represents the flood duration (s).

#### 3. Results

#### 3.1. Water Supply Rates

Figure 2 shows the mean water supply rates from area-volume changes before GLOFs between 2009 and 2012. The mean supply rates ranged between 4.0 and 60.0  $\text{m}^3 \text{ s}^{-1}$  (Figure 2). The maximum mean supply

# **Geophysical Research Letters**



Figure 2. Mean water supply rates (Q<sub>i</sub>) from 2009 to 2012 derived from formula (4).

rate before the GLOF derived from volume changes in 2008 was  $30.9 \text{ m}^3 \text{ s}^{-1}$ , versus rates of 41 to  $60 \text{ m}^3 \text{ s}^{-1}$  in the other years (Table 1). The supply rate during a GLOF could not be measured by using area-volume changes due to the opened englacial channels. Instead, we calculated the supply rate during a GLOF using the difference between flood volumes observed at Xiehela Station and initial lake volumes. The supply rates before a GLOF and during a GLOF were less than 70 m<sup>3</sup> s<sup>-1</sup> from 2009 to 2012. However, the supply rate during the 2008 and 2015 GLOFs was larger than 200 m<sup>3</sup> s<sup>-1</sup>, which is roughly 7 times that before the GLOF and between 3 and 4 times the maximum supply rate from 2009 to 2012. Thus, it is important to investigate why the supply rate during the 2008 and 2015 GLOFs was so large. These results suggest that other components contributed to the supply rate during the GLOFs.

#### 3.2. Two Components of the Flood Volume Supply

Glacial meltwater and icefall (calving) are likely to be the two main components of the flood volume supply (Georgiou et al., 2009; Liu & Fukushima, 1999). To estimate variations in the water supply rate induced by temperature, we used the temperature index and degree-day approaches (Text S1), with data recorded at Tian Shan Station and reanalysis data as input parameters (Figures S1 and S4), since reanalysis data can make up for the deficiencies from Tianshan Station (Kalnay et al., 1996; Ng et al., 2007).

Table 1           Water Supply Rate (Q <sub>i</sub> ) and Flood Volume Derived From the Lake Volume Changes and Observations at Xiehela Station						
			Water supply rate $(Q_i, m^3 s^{-1})$			Flood volume ( $\times 10^8 \text{ m}^3$ )
	Lake area (km <sup>2</sup> )	Initial lake volume (×10 <sup>8</sup> m <sup>3</sup> )	Maximum before the GLOF, derived from lake volume changes	During the GLOF, derived from the difference between flood volume at Xiehela Station and the initial lake volume	Duration (days)	Observation at Xiehela Station
2008	3.50	0.99	30.9	236.1-295.1	4–5	2.01/2.07 <sup>a</sup>
2009	3.33	0.92	48.2	68.0	<8	1.39
2010	3.83	1.12	41.7	47.6	<9	1.49
2011	3.21	0.87	60.0	60.2	5	1.13
2012	2.41	0.57	41.5	46.3	<7	0.85
2015 <sup>b</sup>	~3.0	0.79	43.1	>300	5	>2.1

<sup>a</sup>The value of 2.01 was observed at Xiehela Station, and the value of 2.07 was simulated by Kingslake and Ng (2013a). <sup>b</sup>The area was about 2.7 km<sup>2</sup> based on data from the China-Brazil Earth Resources Satellite (CBERS-04) on 22 July 2015. Based on data from the present study, we estimated that the area was about 3.0 km<sup>2</sup> at the beginning of the GLOF.

The results of this analysis suggested that the number of degree-days of positive temperatures after peak discharge were less than the number of degree-days of positive temperatures for the same number of days before the peak discharge in 1999, 2005, 2009, 2012, and 2015, even when we applied a 2 day time lag for the runoff from the lake to reach Xiehela Station (Table S3). The number of degree-days of positive temperatures for the same number of days before the peak discharge was slightly larger than the number of degree-days of positive temperature for the same number of days before the peak discharge in 2010 and 2011, although the difference was less than 3.0°C. Hence, according to our temperature-index model and our degree-day model (Text S1), the glacial meltwater supplied to the lake before the peak discharge was not larger than the water supplied to the lake after the peak discharge. Even if the temperature changed by 1.0 to  $3.0^{\circ}$ C, the supply rate changed by a maximum of 27.3 m<sup>3</sup> s<sup>-1</sup>. The maximum water supply rates were 60.0 and 41.5 m<sup>3</sup> s<sup>-1</sup> in 2011 and 2012, respectively (Table 1). Hence, although we cannot get the exact supply rate, we can estimate the magnitude of the supply rate with less than a maximum of 87.3 m<sup>3</sup> s<sup>-1</sup>.

Icefalls have been observed at the beginning of the filling of the lake, and the area of floating ice increased slightly during a GLOF (Xie et al., 2013) due to the ice dynamics, which added about  $0.12 \times 10^8$  m<sup>3</sup> ice (7% of the estimated maximum lake volume) (Mayer et al., 2008). This indicates that the water-equivalent calving flux occurred mostly before a GLOF. Consequently, the water supply rate due to the calving flux did not vary greatly because the volume due to the water-equivalent calving flux was part of the initial lake volume in our calculations.

#### 3.3. Water Storage by Glaciers

Water storage by glaciers includes short-, intermediate-, and long-term storage, which result from runoff within mountainous drainage basins at different time scales (Huss, 2011; Jansson et al., 2003). Long-term storage releases water from melting snow and ice in response to climate change. Intermediate-term storage reflects runoff characteristics such as internal water storage and subglacial storage, whereas short-term storage includes singular storage events that result from changes in melting conditions (Jansson et al., 2003). For example, englacial lake outburst floods and glacier surges commonly lead to long-term buildup of water storage followed by a short-term release. However, short-term and intermediate-term glacial storage have not been reported to be causes for outburst floods.

In this study, we derived short-term and intermediate-term water storage by glaciers from discharge records at Xiehela Station. There were two peak flood discharges in both 1999 and 2005 (Figures 3 and S3). In 1999, the first peak flood (1,940 m<sup>3</sup> s<sup>-1</sup>) occurred on 19 July 1999, with a flood volume of about 0.171 km<sup>3</sup>, followed by a second peak flood (2,100 m<sup>3</sup> s<sup>-1</sup>) on 31 July 1999, with a flood volume of about 0.187 km<sup>3</sup> (Figure 3a). It was observed from Landsat ETM+ images (acquired on 25 July 1999) that Lake Merzbacher was filled with water; thus, the lake must have filled again between 20 and 29 July, after the first peak discharge, and then drained again. The average water supply rate of the second flood was 200 to 250 m<sup>3</sup> s<sup>-1</sup>, which was like that in 15 July 2008. However, the temperature decreased by about 2°C during the period around the two floods (Figure 3b). After the second flood, the observed discharge at Xiehela Station remained larger than 1,000 m<sup>3</sup> s<sup>-1</sup> for several days. We also noted that after 13 August 1999, the discharge at Xiehela Station decreased quickly to a value near 350 m<sup>3</sup> s<sup>-1</sup>, which is close to the base flow before the 19 July GLOF.

In 2005, two GLOFs events were also observed (Hagg et al., 2008), but the second peak discharge (1,040 m<sup>3</sup> s<sup>-1</sup> on 4 August) was smaller than the first one (1,840 m<sup>3</sup> s<sup>-1</sup> on 15 July) (Figure S3). The change in water level derived from field observations in 2005 was 0.5 m per day, after which a smaller jökulhlaup occurred (Mayr et al., 2014). In this second case, the water supply rate was less than 24 m<sup>3</sup> s<sup>-1</sup> based on the assumption that the lake area was 4 km<sup>2</sup>. However, the 2005 supply rate was similar to those in 2009 and 2010.

Therefore, the supply rate during the 2008 and 1999 GLOFs differed greatly from that during the GLOFs in 2005, 2009, and 2010. Sources other than glacial meltwater and icefall, such as intermediate-term and short-term water storage by glaciers, must have contributed to the flood volumes in 1999 and 2008.

#### 3.4. Uncertainties of the Initial Lake Volume, the Water Supply Rate, and the Flood Volume

The uncertainty of the lake area extraction was about 7% (Xie et al., 2013). The uncertainty of the lake volume derived from the relationship between lake area and volume (combined with the uncertainties in the DEMs)



**Figure 3.** (a) Flooding in 1999 based on records at Xiehela Station, showing two peak floods. (b) Temperature records derived from the NCEP/NCAR reanalysis data set around the times of the two floods.

was about 2%. This, in turn, created an uncertainty of less than 8% in the water supply rate before the GLOFs. The uncertainty in the water supply rate during GLOFs came mainly from flood volume calculated by the graphical separation methods and from the initial lake volume; however, the HJ satellite passes over our study area at 2 day intervals, the images were acquired before the GLOFs. This will lead to estimation of a water supply rate during GLOFs that is lower than the real rate. Unfortunately, the magnitude of this error cannot be estimated based on our current knowledge. The initial lake area and volume in 2008 and 2011 were relatively accurate because the GLOFs occurred very close to when the satellite images were acquired.

#### 4. Discussion and Conclusion

The combination of remotely sensed data of KH-9, SRTM, and SPOT-5/ALOS DEMs provided a good source for estimation of the storage capacity of Lake Merzbacher in 2000 and 2008. Based on the DEMs, we constructed a relationship between the lake's area and volume and used the relationship to derive the lake's water supply rate. In this study, we directly used the change in lake volume to calculate the water supply rate which has the advantage to include the effects of glacial meltwater, precipitation, and calving of ice. Our results showed that the mean water supply rates were 40–60 m<sup>3</sup> s<sup>-1</sup> with a maximum of 87.3 m<sup>3</sup> s<sup>-1</sup> due to climate and ice-fall, which is agreement on ~31 to 52 m<sup>3</sup> s<sup>-1</sup> owing to glacier melting during summer (Mavlyudov, 1995). In addition, we also obtained some large mean supply rates at 200–300 m<sup>3</sup> s<sup>-1</sup> during GLOFs, such as in 1999, 2008, and 2015.

The GLOFs from Lake Merzbacher were triggered by a gradual increase in the summer air temperature (Ng et al., 2007) that affected melting of the ice lake dam created by the southern Inylchek Glacier (Wortmann et al., 2014). Although there is lack of sufficient precipitation data for a thorough analysis, we were able to obtain precipitation data from the Tianshan Station (78.2°N, 41.9°E, 3614 m asl; Table S2). The precipitation before and after the peak flood were less than 15 mm in all years except 1999. Between the two peaks in 1999, the precipitation was 8.8 mm. The precipitation was about 1 mm during the 2008 GLOF, of which the low amount suggests that the precipitation had little effect on the GLOFs. As a result, we ruled out variability in the water supply rate due to variability in precipitation as a major factor.

The northern Inylchek Glacier surged 3.5 km in late 1996 and early 1997 (Häusler et al., 2011, 2016). Although there were lack of satellite images from that time, a high peak discharge (about 825 m<sup>3</sup> s<sup>-1</sup>) observed at Xiehela Station on 5 December 1996 agreed with the reported surge dates. Regardless of whether the glacier surge triggered the 1996/1997 GLOF, the flood volume (0.284 km<sup>3</sup>) was found to be more than the maximum lake volume storage (0.120 km<sup>3</sup>). The total precipitation and mean daily temperature were 2 mm and  $-20.2^{\circ}$ C at Tianshan Station in December 1996. This low precipitation suggests that the change in the water supply caused by climatic factors was less important than other factors. In contrast, glacier movement would lead to englacial melting and release of englacial water storage (Bartholomaus et al., 2008; Burgess et al., 2013; Lingle & Fatland, 2003; Mayer et al., 2008). Therefore, the observed supply rate should be resulted from more than meteorological factors (melting due to temperature increases, precipitation) and icefall (calving); it must also have been resulted from changes in englacial water storage.

The key finding from the present study is that the supply rates in 2008 and 2015 (larger than 200 m<sup>3</sup> s<sup>-1</sup> in both years) differed from those in 2009 and 2012 (less than 70 m<sup>3</sup> s<sup>-1</sup>). The events in 1999 and 2005 were also different in having double peaks in both years. Furthermore, glacier surges in 1996 released more than 0.284 km<sup>3</sup> water, which were recorded by the flood discharges observed at Xiehela Station. Because of the limited availability of remote sensing data, we can only analyze GLOFs in several representative years. Although our hypothesis about the water source is based on relatively limited data, our results suggest the importance of englacial water release during GLOFs. Therefore, we believe that both short-term and intermediate-term storage water contributed to the observed GLOFs from Lake Merzbacher, in addition to the contributions of glacial meltwater and icefall (calving) to certain flood events. However, we need further investigation to verify this finding in future. The irregular frequency of releases of short-term and intermediate-term storage water may explain why the outburst floods are unpredictable. At present, it is hard for us to simulate the volume of the englacial water release when the flood volume also includes this additional form of water release. In addition, we cannot determine the conditions under which the sudden floods are accompanied by a release of englacial water. Future studies are needed to improve these issues.

Based on our results, we can categorize the GLOF events into three general distinct types: case I (singular event-triggered englacial water release), case II (glacier melt due to temperature changes), and case III (englacial water release mixed with glacier melt) in Figure S5. In case I, the flood volume in 1996 resulted from a combination of glacier surges and the release of englacial water storage along with icefall, precipitation, and glacier melting. In case II, for the flood events in 2005, 2009, and 2010, the flood volume observed by remote sensing agreed with observations at Xiehela Station. In this case, the flood volume mainly resulted from melting of the glacier caused by high temperatures. In case III, the flood volumes in 1999, 2008, and 2015 were more complicated to estimate: the response of glacial meltwater to climate change (e.g., temperature rise) is one reason, but the release of short-term and intermediate-term storage water by the glacier was another. Both cases (I and III) included the release of stored englacial water and caused more irregular GLOFs with large flood volumes that are difficult to predict. Therefore, it will be necessary to improve our understanding of the processes that occur within and under the glacier before we can better predict their effects on GLOFs.

### References

Aizen, V. B., Aizen, E. M., Melack, J. M., & Dozier, J. (1997). Climatic and hydrologic changes in the Tienshan, Central Asia. Journal of Climate, 10, 12. https://doi.org/10.1175/1520-0442(1997)010%3C1393:CAHCIT%3E2.0.CO;2

Bartholomaus, T. C., Anderson, R. S., & Anderson, S. P. (2008). Response of glacier basal motion to transient water storage. *Nature Geoscience*, 1(1), 33–37. https://doi.org/10.1038/ngeo.2007.52

Berthier, E., & Toutin, T. (2008). SPOT5\_HRS digital elevation models and the monitoring of glacier elevation changes in North-West Canada and South-East Alaska. *Remote Sensing of Environment*, *112*, 2443–2454. https://doi.org/10.1016/j.rse.2007.11.004

Bolch, T., Peters, J., Yegorov, A., Pradhan, B., Buchroithner, M., & Blagoveshchensky, V. (2011). Identification of potentially dangerous glacial lakes in the northern Tien Shan. *Natural Hazards*, 59(3), 1691–1714. https://doi.org/10.1007/s11069-011-9860-2

Burgess, E. W., Forster, R. R., & Larsen, C. F. (2013). Flow velocities of Alaskan glaciers. Nature Communications, 4, 2146. https://doi.org/ 10.1038/ncomms3146

Fan, Y., Chen, Y., Liu, Y., & Li, W. (2013). Variation of baseflows in the headstreams of the Tarim River Basin during 1960–2007. Journal of Hydrology, 487, 98–108. https://doi.org/10.1016/j.jhydrol.2013.02.037

Fujita, K., Sakai, A., Takenaka, S., Nuimura, T., Surazakov, A. B., Sawagaki, T., & Yamanokuchi, T. (2013). Potential flood volume of Himalayan glacial lakes. *Nature Hazards and Earth System Sciences*, *13*, 1827–1839. https://doi.org/10.5194/nhess-13-1827-2013

Georgiou, S., Shepherd, A., Mcmillan, M., & Nienow, P. (2009). Seasonal evolution of supraglacial lake volume from ASTER imagery. Annals of Glaciology, 50(52), 95–100.

Glazirin, G. E. (2010). A century of investigations on outbursts of the ice-dammed lake Merzbacher(central Tien Shan). Austrian Journal of Earth Sciences, 103(2), 171–178.

#### Acknowledgments

We thank Geoff Hart and Anjum Muhammad Naveed for reviewing the English in an early version of this manuscript. Special appreciation is due to Tobias Bolch for helpful suggestions on this paper. This work was supported by the Ministry of Science and Technology of the People's Republic of China (grant: 2013FY111400, 2013CBA01808, and 2013BAC10B01), the National Natural Science Foundation of China (grant: 41671066 and 41271082), the Chinese Academy of Sciences (HHS-TSS-STS-1501 and SkLCS-ZZ-2017), and the China Scholarship Council. We thank USGS for providing the Landsat data (http://glovis.usgs. gov/), China Centre for Resources Satellite Data and Application for providing HJ 1A/1B (http://www.cresda. com/), ESRL for NCEP/NCAR Reanalysis data (https://www.esrl.noaa.gov/psd/ data/gridded/data.ncep.reanalysis.html) in this study.

- Hagg, W., Mayer, C., Lambrecht, A., Helm, A., & Michailjow, W. (2008). Glaciological results of the 2005 expedition to Inylchek glacier, Central Tian Shan. *Geography, Environment, Sustainability*, 1, 38–45.
- Häusler, H., Ng, F., Kopecny, A., & Leber, D. (2016). Remote-sensing-based analysis of the 1996 surge of Northern Inylchek Glacier, central Tien Shan, Kyrqyzstan. *Geomorphology*, 273, 292–307. https://doi.org/10.1016/j.geomorph.2016.08.021

Häusler, H., Scheibz, J., Leber, B., Kopecny, A., Echtler, H., Wetzel, H.-U., & Moldobekov, B. (2011). Results from the 2009 geoscientific expedition to the Inylchek glacier, Central Tien Shan (Kyrgyzstan). Austrian Journal of Earth Sciences, 104(2), 47–57.

Huggel, C., Kääb, A., Haebeli, W., Teysseire, P., & Paul, F. (2002). Remote sensing based assessment of hazards from glacier lake outburst: A case study in the Swiss Alpis. *Canadian Geotechnical Journal*, *39*, 316–330. https://doi.org/10.1139/T01-099

Huss, M. (2011). Present and future contribution of glacier storage change to runoff from macroscale drainage basins in Europe. *Water Resources Research*, *47*, W07511. https://doi.org/10.1029/2010WR010299

- Iwata, S., Ageta, Y., Naito, N., Sakai, A., Narama, C., & Karma (2002). Glacial lakes and their outburst flood assessment in the Bhutan Himalaya. Global Environmental Research, 6(1), 3–17.
- Janský, B., Šobr, M., & Engel, Z. (2010). Outburst flood hazard: Case studies from the Tien-Shan Mountains, Kyrgyzstan. Limnologica Ecology and Management of Inland Waters, 40(4), 358–364. https://doi.org/10.1016/j.limno.2009.11.013

Jansson, P., Hock, R., & Schneider, T. (2003). The concept of glacier storage: A review. Journal of Hydrology, 282(1–4), 116–129. https://doi.org/ 10.1016/S0022-1694(03)00258-0

Jongman, B., Hochrainer-Stigler, S., Feyen, L., Aerts, J. C. J. H., Mechler, R., Botzen, W. J. W., ... Ward, P. J. (2014). Increasing stress on disaster-risk finance due to large floods. *Nature Climate Change*, 4(4), 264–268. https://doi.org/10.1038/nclimate2124

Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., ... Joseph, D. (1996). The NCEP/NCAR 40-year reanalysis project. *Bulletin of the American Meteorological Society*, *77*, 437–470. https://doi.org/10.1175/1520-0477(1996)077%3C0437:TNYRP%3E2.0.CO;2

- Kingslake, J., & Ng, F. (2013a). Modelling the coupling of flood discharge with glacier flow during Jokulhlaups. Annals of Glaciology, 54(63), 25–31. https://doi.org/10.3189/2013AoG63A331
- Kingslake, J., & Ng, F. (2013b). Quantifying the predictability of the timing of Jokulhlaups from Merzbacher Lake, Kyrgyzstan. Journal of Glaciology, 59(217), 805–818. https://doi.org/10.3189/2013JoG12J156
- Konovalov, A. G. (1991), Methods for computing the onset date and daily discharge hydrograph of the outburst from Mertzbacher Lake, Northern Inylchek Glacier, Tien Shan, Glaciers-ocean-atmosphere interactions. IAHS Publ. No. 208, pp. 359–366.
- Kundzewicz, Z. W., Merz, B., Vorogushyn, S., Hartmann, H., Duethmann, D., Wortmann, M., ... Krysanova, V. (2015). Analysis of changes in climate and river discharge with focus on seasonal runoff predictability in the Aksu River Basin. *Environmental Earth Sciences*, 73, 501–516. https://doi.org/10.1007/s12665-014-3137-5
- Kuzmichenok, V. A. (1984). K voprosu ob obrabotke materialov topograficheskikh s' emok dna glyatsial'nykh ozer[Questioning the treatment of materials from topographical surveys of glacial lake beds], Gliatsiologicheskie Issledova-niya v Tsentral' nom Tyan-Shane (Glaciological studies in Central Tien Shan), pp. 130–147.

Lingle, C. S., & Fatland, D. R. (2003). Does englacial water storage drive temperate glacier surges? *Annals of Glaciology*, *36*, 14–20. Liu, J. (1992). Jokulhlaups in the Kunmalike River, southern Tien Shan mountains, China. *Annals of Glaciology*, *16*, 85–88.

Liu, J., & Fukushima, Y. (1999). Recent change and prediction of glacier-dammed lake outburst floods from Kunmalik River in southern Tien Shan, China., IAHS Publ., 255 ((Symposium at Birmingham, UK 1999 – Hydrological Extremes: Understanding, Predicting, Mitigating)), pp. 99–107.

Mavlyudov, B. R. (1995). Tongue oscillations of Northern Inylchek Glacier (in Russian with English summary). Data Glacier Studies, 79, 95–98.
Mayer, C., Lambrecht, A., Hagg, W., Helm, A., & Scharrer, K. (2008). Post-drainage ice dam response at Lake Merzbacher, Inylchek Glacier, Kyrgyzstan. Geografiska Annaler. Series A, Physical Geography, 90(1), 87–96. https://doi.org/10.1111/j.1468-0459.2008.00336.x

- Mayr, E., Juen, M., Mayer, C., Usubaliev, R., & Hagg, W. (2014). Modeling runoff from the Inylchek glaciers and filling of ice-dammed lake Merzbacher, Central Tian Shan. *Geofrafiska Annaler Series A: Physical Geography*, 96(4), 1–16. https://doi.org/10.1111/geoa.12061
- Merzbacher, G. (1905). The central Tian-shan Mountains 1902–1903 (pp. 199–202). London, London: John Murray.
- Ng, F., & Liu, S. (2009). Temporal dynamics of a Jokulhlaup system. Journal of Glaciology, 55(192), 651–665. https://doi.org/10.3189/ 002214309789470897
- Ng, F., Liu, S., Mavlyudov, B., & Wang, Y. (2007). Climatic control on the peak discharge of glacier outburst floods. *Geophysical Research Letters*, 34, L21503. https://doi.org/10.1029/2007GL031426

Nuth, C., & Kääb, A. (2011). Co-registration and bias corrections of satellite elevation data sets for quantifying glacier thickness change. *The Cryosphere*, *5*(1), 271–290. https://doi.org/10.5194/tc-5-271-2011

- Osti, R., Egashira, S., & Adikari, Y. (2013). Prediction and assessment of multiple glacial lake outburst floods scenario in Pho Chu River basin, Bhutan. *Hydrological Processes*, 27(2), 262–274. https://doi.org/10.1002/hyp.8342
- Richardson, S. D., & Reynolds, J. M. (2000). An overview of glacial hazards in the Himalayas. Quaternary International, 65–66, 31–47. https://doi.org/10.1016/S1040-6182(99)00035-X
- Shangguan, D., Bolch, T., Ding, Y., Kröhnert, M., Pieczonka, T., Wetzel, H. U., & Liu, S. (2015). Mass changes of Southern and Northern Inylchek Glacier, Central Tian Shan, Kyrgyzstan, during ~1975 and 2007 derived from remote sensing data. *The Cryosphere*, 9(2), 703–717, https:// doi.org/10.5194/tc-9-703-2015
- Shen, Y., Wang, G., Ding, Y., Su, H., Mao, W., Wang, S., & Duishen, M. M. (2009). Changes in Merzbacher Lake of Inylchek Glacier and glacial flash floods in Aksu River Basin, Tianshan during the period of 1903–2009. *Journal of Glaciology and Geocryology*, 31(6), 993–1002.
- Sorg, A., Bolch, T., Stoffel, M., Solomina, O., & Beniston, M. (2012). Climate change impacts on glaciers and runoff in Tien Shan (Central Asia). *Nature Climate Change*, 2, 725–731. https://doi.org/10.1038/nclimate1592
- Uchiyama, Y., Honda, M., Mizuta, Y., Otsuka, K., Ishizeki, T., Okatani, T., & Tamura, E. (2008). Revising 1: 25,000-scale topographic maps using ALOS/PRISM Imagery. *Bulletin of the Geographical Survey Institute*, *56*, 1–15.
- Wang, X., Ding, Y., Liu, S., Jiang, L., Wu, K., Z. Jiang, & Guo, W. (2013). Changes of glacial lakes and implications in Tian Shan, central Asia, based on remote sensing data from 1990 to 2010. *Environmental Research Letters*, 8(4), 044052. https://doi.org/10.1088/1748-9326/8/4/ 044052
- Wortmann, M., Krysanova, V., Kundzewicz, Z. W., Su, B., & Li, X. (2014). Assessing the influence of the Merzbacher Lake outburst floods on discharge using the hydrological model SWIM in the Aksu headwaters, Kyrgyzstan/NW China. *Hydrological Processes*, *28*(26), 6337–6350. https://doi.org/10.1002/hyp.10118
- Xie, Z., ShangGuan, D., Zhang, S., Ding, Y., & Liu, S. (2013). Index for hazard of Glacier Lake Outburst flood of Lake Merzbacher by satellite-based monitoring of lake area and ice cover. *Global and Planetary Change*, *107*, 229–237. https://doi.org/10.1016/j.gloplacha.2012.05.025

- Yao, T., Li, Z., Yang, W., Guo, X., Zhu, L., Kang, S., ... Yu, W. (2010). Glacial distribution and mass balance in the Yarlung Zangbo River and its influence on lakes. *Chinese Science Bulletin*, 55(20), 2072–2078. https://doi.org/10.1007/s11434-010-3213-5
- Yao, X., Liu, S., Sun, M., Wei, J., & Guo, W. (2012). Volume calculation and analysis of the changes in moriane-dammed lakes in the north Himalaya: A case study of Longbasaba lake. *Journal of Glaciology*, 58(210), 753–760. https://doi.org/10.3189/ 2012JoG11J048
  - Zhang, G., Yao, T., Shum, C. K., Yi, S., Yang, K., Xie, H., ... Yu, J. (2017). Lake volume and groundwater storage variations in Tibetan Plateau's endorheic basin. *Geophysical Research Letters*, 44, 5550–5560. https://doi.org/10.1002/2017GL073773
  - Zhang, G., Yao, T., Xie, H., Wang, W., & Yang, W. (2015). An inventory of glacial lakes in the Third Pole region and their changes in response to global warming. *Global and Planetary Change*, 131, 148–157. https://doi.org/10.1016/j.gloplacha.2015.05.013